Author's response to reviews

Title: Predictive Value of Coronary Calcifications for Future Cardiac Events in Asymptomatic Patients with Diabetes mellitus: A prospective Study in 716 Patients over 8 Years

Authors:

Alexander Becker (alexander.becker@med.uni-muenchen.de)
Alexander W Leber (alexander.leber@med.uni-muenchen.de)
Christoph Becker (christoph.becker@med.uni-muenchen.de)
Franz von Ziegler (franz.ziegler@med.uni-muenchen.de)
Janine Tittus (janine.tittus@med.uni-muenchen.de)
Ines Schroeder (ines.schroeder@med.uni-muenchen.de)
Gerhard Steinbeck (gerhard.steinbeck@med.uni-muenchen.de)
Andreas Knez (andreas.knez@med.uni-muenchen.de)

Version: 2 Date: 28 May 2008

Author's response to reviews: see over
Submitting Original Article to BMC Cardiovascular Disorders.

Title: Predictive Value of Coronary Calcifications for Future Cardiac Events in Asymptomatic Patients with Diabetes mellitus: A prospective Study in 716 Patients of 8 Years


Dear Editor,
we would like to submit the above-cited manuscript for consideration for publication in BMC Cardiovascular Disorders.

This prospective study in an asymptomatic population of patients with diabetes mellitus depicts the predictive value of coronary calcifications for future cardiovascular events. It is the first study with a long-term follow up of 8 years. We feel that this findings will be of interest to your readers and look forward to your decision.

All coauthors have seen and agreed to the content of the manuscript. No author has a financial interest in the subject matter or materials mentioned in the paper. A. Becker and A. Knez are responsible for clinical trial data.

With best regards,

A. Becker MD

Alexander Becker, MD
Department of Cardiology
Ludwig-Maximilians-University, Munich
D-81377 Munich
Germany
Tel: +49 89 7095-2229
Fax: +49 89 7095-2227
Email: alexander.becker@med.uni-muenchen.de